Author's response to reviews

Title: Clinical outcome of pneumococcal meningitis during the emergence of pencillin-resistant Streptococcus pneumoniae: an observational study

Authors:

Edilane L Gouveia (edilane@gmail.com)
Joice N Reis (joice@ufba.br)
Brendan Flannery (bflannery@cdc.gov)
Soraia M Cordeiro (soraiamc@yahoo.com.br)
Josilene BT Lima (josilene_lima@yahoo.com.br)
Ricardo M Pinheiro (rico_pinheiro@hotmail.com)
Katia Salgado (katia_salgado@yahoo.com.br)
Ana Veronica Mascarenhas (mascarenhasav@hotmail.com)
Gloria S Carvalho (msc8@cdc.gov)
Bernie Beall (beb0@cdc.gov)
Mitermayer G Reis (miter@bahia.fiocruz.br)
Albert I Ko (albert.ko@yale.edu)

Version: 6 Date: 8 November 2011

Author's response to reviews:

We have corrected the Author list in the manuscript so that authors are listed exactly as they are in the submission system (changes were made to Gloria S Carvalho) and Bernard W Beall was corrected in the submission system.